\documentclass[11pt]{article}

% This package contains many of the macros you may need.
%
% A useful environment defined in it is the example environment, which you can
% invoke via \begin{example} and \end{example}.
\usepackage{scribe}

\begin{document}

\begin{center}
  \fcolorbox{black}{white}{
    \begin{minipage}{.9\columnwidth}
      \textbf{Stats 300b: Theory of Statistics}
      \hfill
      \textbf{Winter 2017} \\
      \vspace{-.2cm}
      % Instead of Lecture 1, January 10, put the actual lecture number and
      % date of the lecture here
      \begin{center}
        {\LARGE Lecture 1 -- January 10} \\
      \end{center}
      \emph{Lecturer: John Duchi}
      \hfill
      \emph{Scribe: }  % You should put your name here if you scribe
    \end{minipage}
  }
\end{center}

\begin{center}
  \dbend ~~  \emph{\textbf{Warning:} these notes may contain factual errors}
\end{center}

\paragraph{Reading:} % Put the reading covered in the lecture here

%%%%%%%%%%%%%%%%% SCRIBE NOTES HERE %%%%%%%%%%%%%%%%%%%%


\end{document}
